Author's response to reviews

Title: Personal Endotoxin Exposure in School Children with Asthma

Authors:

    Ralph J Delfino (rdelfino@uci.edu)
    Norbert Staimer (nstaimer@uci.edu)
    Thomas Tjoa (ttjoa@uci.edu)

Version: 4 Date: 1 August 2011

Author's response to reviews: see over
Dear Editor,

Please find enclosed the above revised manuscript for submission to *Environmental Health* as an original research article. The Editor’s requested edits have been made, namely:

Formatting changes

References: that were not cited in order were corrected.

Tables: table 2 and 5 different parts have been merged into a single table.

Tracking: all forms of tracking were remove in the manuscript.

Typography: The manuscript was checked and corrected for a few typographical and grammatical errors.

Sincerely,

Ralph J. Delfino, MD, PhD